Author's response to reviews

Title: The relationship between mentoring on healthy behaviors and well-being among Israeli youth in boarding schools: A mixed-methods study

Authors:

Maayan Agmon (agmon.mn@gmail.com)
Cheryl Zlotnick (czlotnick@univ.haifa.ac.il)
Anat Finkelstein (fanat@zahav.net.il)

Version: 3 Date: 27 July 2014

Author's response to reviews:

July 27, 2014

Dear Editor:

Thank you for noticing important points for change in the manuscript.

According to your request, we have replaced Table 1. With a new table, presenting participants' information in aggregated form.

Additionally, we would like to replace our second recommended peer reviewer with:

Jean E. Rhodes
University of Massachusetts Boston
Jean.Rhodes@umb.edu

Sincerely yours,

Maayan Agmon.